Author's response to reviews

Title: The Impact of Direct Provision Accommodation for Asylum Seekers on Organisation and Delivery of Local Primary Care and Social Care Services: A Case Study

Authors:

Hans-Olaf Pieper (hopieper@hotmail.com)
Pauline Clerkin (pauline.clerkin@nuigalway.ie)
Anne MacFarlane (anne.macfarlane@nuigalway.ie)

Version: 5 Date: 15 April 2011

Author's response to reviews:

April 15 2011

Dear Professor Harris,

RE: MS: 3116265374630609

Thank you for your support with the publication of this manuscript.

As per your last e-mail we have made some final changes to the manuscript:

We have explained the term "direct provision accommodation" in the abstract.

We have gone through the manuscript formatting checklist one more time and we have ensured that this revised manuscript conforms to all of the points. The following changes have been made as part of that review:

We have removed an asterix that appeared after first and last authors' names.

We have inserted the word [sic] after the use of the term “coloured” skin in the Results section 3 because we wish to acknowledge that this is a participant’s term rather than our own.

We have converted Figure 1 from colour to black and white and updated the URL address (reference 21) that goes with this figure.

There have been no other changes and we are happy with this as the final version of our manuscript.

Yours sincerely,

Anne MacFarlane
Lecturer in Primary Care
Discipline of General Practice
NUI Galway
Galway
Ireland